Author's response to reviews

Title: Cognitive training plus a comprehensive psychosocial programme (OPUS) versus the comprehensive psychosocial programme alone for patients with first-episode schizophrenia (the NEUROCOM trial): A study protocol for a centrally randomised, observer-blinded multi-centre clinical trial

Authors:

Lone Vesterager (lone.vesterager@regionh.dk)
Torben Christensen (torbchri@rm.dk)
Birthe Olsen (birtolse@rm.dk)
Gertrud Krarup (gertkra@rm.dk)
Hysse Forchhammer (hbf@glo.regionh.dk)
Marianne Melau (marianne.melau@regionh.dk)
Christian Gluud (cgluud@ctu.rh.dk)
Merete Nordentoft (mn@dclnet.dk)

Version: 4 Date: 28 January 2011

Author's response to reviews:

full ref. 31 is provided
Table 1 is formatted using a tables tool and moved to the manuscript below the references
middlenames/initials are corrected